Author's response to reviews

Title: Combination Strategies for Pandemic Influenza Response - a Systematic Review of Mathematical Modeling Studies

Authors:

Vernon Lee (vernonljm@hotmail.com)
David C Lye (David_LYE@ttsh.com.sg)
Annelies Wilder-Smith (epwvs@pacific.net.sg)

Version: 3 Date: 20 November 2009

Author's response to reviews:

Dear Dr Cassady-Cain,

We wish to thank you and the reviewers for the kind comments to our revisions.

With regards to Dr Carl Heneghan’s comment, we agree that there are indeed other databases in existence, and also other unpublished (or yet to be published) material that may be missed. We have indicated this limitation in the discussion section.

We sincerely hope that we have covered most of the relevant articles around this topic, and hope that the study will be useful for clinicians, public health workers, and policy makers in planning for the current pandemic, and for future research prioritization.

Regards,
Vernon Lee
On behalf of the authors